Author's response to reviews

Title: Reproducibility of wrist home blood pressure measurement with position sensor and automatic data storage

Authors:

Sakir Uen (suen@uni-bonn.de)
Rolf Fimmers (Rolf.Fimmers@ukb.uni-bonn.de)
Miriam Brieger (miriam.brieger@gmx.de)
Georg Nickenig (Georg.Nickenig@ukb.uni-bonn.de)
Thomas Mengden (mengden@uni-bonn.de)

Version: 4 Date: 15 December 2008

Author's response to reviews:

The abstract was structured according to the guidelines provided from BMC. An acknowledgments section was included to the manuscript.